Author's response to reviews

Title: Addison's disease presenting as idiopathic intracranial hypertension

Authors:

DUSHYANT SHARMA (dushyantsharma@hotmail.com)
ROHINI MUKHERJEE (rohini.mukherjee@aintree.nhs.uk)
PETER MOORE (peter.moore@thewaltoncentre.nhs.uk)
DANIEL J CUTHBERTSON (Daniel.Cuthbertson@liverpool.ac.uk)

Version: 3 Date: 28 June 2008

Author's response to reviews:

Thanks for suggesting the changes in manuscript which I have tried to make and are as follows:

Title has been restuctured
e mail Ids have been added in the author title page
Abstract has been put on page 2 with structure suggested
case report has been rephrased as case presentation
patients ethnic origin has been added
conclusion has been added
list of abbreviations added
competing interest section added
authors contribution section added
acknowledgement added
consent section added
figure title added
tried to conceal patients identity by increasing the black area around eyes
references restructured
hope this is up to your expectation